Author’s response to reviews

Title: Knowledge and attitudes of primary health care physicians and nurses with regard to population screening for colorectal cancer in Balearic Islands and Barcelona.

Authors:

Maria Ramos (mramos@dgsanita.caib.es)
Magdalena Esteva (mesteva@ibsalut.caib.es)
Jesus Almeda (jalmeda@ambitcp.catsalut.net)
Elena Cabeza (ecabeza@dgsanita.caib.es)
Diana Puente (dpuente@idiapigol.org)
Rosa Saladich (rsaladich@gmail.com)
Albert Boada (aboadav.bcn.ics@gencat.cat)
Maria Llagostera (mllagostera.cp.ics@gencat.cat)

Version: 6 Date: 31 August 2010

Author’s response to reviews: see over
Srs. Editors BMC Cancer:

I submit you the manuscript “Knowledge and attitudes of primary health care physicians and nurses with regard to population screening for colorectal cancer” reviewed according referees’ suggestions.

Maria Ramos

Answers to your third comments are detailed below:

Referee Monika Janda
- We have changed “participate in a screening program” by “support a screening program”. We are grateful for this comment because the text is now more clear.
- We did our best for improving the discussion, trying to compare our results with the previous literature.
- We have reviewed the last paragraph of the discussion section, where there are not new data not presented in the results. This paragraph describes and comments the results of the multivariate analysis (Table 4).

Referee Paolo Giorgi Rossi
- We have explicated in the methods statistical analysis that we have not taken into account the sample structure in the analysis.